



# Article 3D Reconstruction of Non-Rigid Plants and Sensor Data Fusion for Agriculture Phenotyping

Gustavo Scalabrini Sampaio<sup>1,\*</sup>, Leandro A. Silva<sup>1</sup> and Maurício Marengoni<sup>2</sup>

- <sup>1</sup> Graduate Program in Electrical Engineering and Computing, Mackenzie Presbyterian University, Rua da Consolação, 896, Prédio 30, Consolação, São Paulo 01302-907, Brazil; leandroaugusto.silva@mackenzie.br
- <sup>2</sup> Department of Computer Science, Federal University of Minas Gerais, Avenida Antônio Carlos, 6627, Prédio do ICEx, Pampulha, Belo Horizonte 31270-901, Brazil; mmarengoni@dcc.ufmg.br
- \* Correspondence: gustavo.sampaio@mackenzista.com.br

**Abstract**: Technology has been promoting a great transformation in farming. The introduction of robotics; the use of sensors in the field; and the advances in computer vision; allow new systems to be developed to assist processes, such as phenotyping, of crop's life cycle monitoring. This work presents, which we believe to be the first time, a system capable of generating 3D models of non-rigid corn plants, which can be used as a tool in the phenotyping process. The system is composed by two modules: an terrestrial acquisition module and a processing module. The terrestrial acquisition module is composed by a robot, equipped with an RGB-D camera and three sets of temperature, humidity, and luminosity sensors, that collects data in the field. The processing module conducts the non-rigid 3D plants reconstruction and merges the sensor data into these models. The work presented here also shows a novel technique for background removal in depth images, as well as efficient techniques for processing these images and the sensor data. Experiments have shown that from the models generated and the data collected, plant structural measurements can be performed accurately and the plant's environment can be mapped, allowing the plant's health to be evaluated and providing greater crop efficiency.

Keywords: 3D reconstruction; computer vision; sensor data fusion; robotics; agriculture

## 1. Introduction

The increased demand for agricultural products, whether to feed the world's population, to supply inputs to the industry or even to trade these products on the commodities market, has encouraged producers to invest in technologies to improve production efficiency. Seeding and harvesting machines are examples of technologies responsible for the agricultural production substantial growth. The most recent figures from the Food and Agriculture Organization of the United Nations, going back to 2019, shows that about 12 billion tons of products from the field are produced annually [1]. This scenario is favorable for research and development of technologies that help the producer; not only at the beginning and ending stages of cultivation, but also at its intermediate phases, helping to identify problems earlier so decisions can be made faster and more accurate [1,2].

The plant growth monitoring is essential to improve crop efficiency and minimize losses. One of the processes that helps this monitoring is phenotyping. The analysis of the plant phenotype, that is, its morphological and physiological characteristics coming from the species genetics and the environmental influence, can be used to evaluate the individual and collective health of plants. Some phenotypic characteristics are: the plant size; the leaves width; the leaves and fruits density; and the angulation between the stems and the leaves; among others. This process and other technologies, such as smart irrigation systems and real-time monitoring, allow us to follow the plant's life cycle, identifying health problems early on; to identify holes in the plantation, due to non-birth or poor



Citation: Sampaio, G.S.; Silva, L.A.; Marengoni, M. 3D Reconstruction of Non-Rigid Plants and Sensor Data Fusion for Agriculture Phenotyping. *Sensors* **2021**, *21*, 4115. https:// doi.org/10.3390/s21124115

Academic Editors: Lapo Governi, Francesco Buonamici and Anabela Oliveira

Received: 29 April 2021 Accepted: 9 June 2021 Published: 15 June 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). formation; and to use fertilizers, pesticides, and water in a rational way. This monitoring aims to increase the efficiency of production, and therefore its volume, and to reduce labor and farm input costs [2–5].

With the research advances in the three-dimensional (3D) reconstruction area, some work has been proposed to use 3D reconstruction of plants in phenotyping processes. Most of these works are limited to controlled environments, avoiding natural factors such as wind and rain that cause movements in plants. Other works, which use field data, make use of static images only. These restrictions are due to the difficulty of generating 3D models of objects with dynamic characteristics, limiting the results seen so far to rigid reconstructions. However, non-rigid reconstruction is important to acquire more information about the objects in the scene, as it allows the accumulation of data from different frames. Moreover, non-rigid reconstruction of plants can be a first step towards modeling the plantation in a complete and continuous way, allowing its global visualization [2,3,6–8].

This work is inserted in this context. Not to perform phenotyping or plant growth analysis, but to provide inputs for these processes. Collecting data in the field and processing them manually is often unfeasible, since the producer would have to travel a great distance, depending on the size of the plantation, at various stages of the plant's life [2,8,9]. Therefore, the objective of this work is to present a system for field data collection and processing to generate accurate 3D models of non-rigid corn plants in an uncontrolled environment, which, we believe, is the first system to do this; and merge this model with sensor data. Specifically, the system was composed of two modules: the acquisition module and the processing module. The acquisition module has aerial and terrestrial subsystems; both subsystems collect data from the field for post-processing; this work will present the development and characteristics of the terrestrial subsystem, called the terrestrial acquisition module from now on. This module is composed of a robot equipped with three sets of temperature, humidity, and luminosity sensors and an RGB-D (Red, Blue, Green and Depth) camera, and is responsible for navigating inside the plantation and collecting the data. In turn, the processing module is responsible for performing non-rigid 3D reconstruction of the plants generating a 3D model with precise measurements, as well as merging the sensors data into these models. The proposed system macro view can be seen in Figure 1. Besides these important contributions, the present work implements a novel technique for background removal in depth images, based on the logarithmic function. Several 3D reconstruction techniques were analyzed, and those that proved to be efficient were selected to be part of the system, such as point cloud treatment, surface reconstruction, and model deformation. The system was developed to work with corn crops, but can be adjusted in the future to meet the needs of other crops.



**Figure 1.** Proposed system macro view. The terrestrial acquisition module collects temperature, humidity, and luminosity data and RGB-D videos; the processing module performs a non-rigid 3D reconstruction of the plants and merge the sensors data into these models. With these modules, the system is able to generate inputs for phenotyping processes.

To present the system developed and the results achieved, this work was organized as follows: Section 2 presents the related works, focusing on 3D reconstruction and similar works that perform 3D reconstruction of plants; Section 3 presents the materials, methods, and techniques used to build the modules that compose the system; Section 4 presents the results, by showing the outputs provided by some tests; Section 5 presents the discussion of the results and proposals for future improvements to the system; Section 6 presents the conclusions.

## 2. Related Works

# 2.1. Robotics in Agriculture

The introduction of large machinery in agriculture, mainly to perform the seeding and harvesting of crops, has contributed significantly to consolidating today's high agricultural production rates. Nowadays, agriculture is beginning to experience the phase of robotization [10–12]. Robotics in agriculture enables and facilitates the functionalities of precision agriculture, characterized by the application of techniques aimed to improve the efficiency of agricultural cultivation and better use and allocation of resources, such as soil, pesticides, and water [13]. In this sense, tasks in agriculture tend to be increasingly more precise with the participation of robots, as well as more sustainable.

From mapping agricultural activities, it can be observed that most of them, especially in the academic context, can already be performed by robots. Traction [14,15], seeding [16–19], detection and mapping [8,9], monitoring and phenotyping [2,20–23], pest control [24–28] and harvesting [29–32] are some activities that have already been addressed in proposing robotic systems for agriculture in different crops [33]. Among these activities, detection and mapping; monitoring and phenotyping; and pest control play a prominent role. These activities are performed in the intermediate phase of the plant's life cycle to follow its growth.

The detection and mapping of plants is an essential activity, as it enables the other activities to be carried out. This task can be performed by robots equipped with geolocation sensors (Global Positioning System-GPS), three-dimensional sensors (Light Detection Furthermore, Ranging (LiDAR) and RGB-D cameras) and Simultaneous Localization and Mapping (SLAM) techniques, as the system proposed by Weiss and Biber (2011) [9] or the system proposed by Potena et al. (2019) [8], which use data coming from a robot and a drone for crop mapping. While detection and mapping can be performed only once during the plant's life cycle, monitoring, phenotyping and pest control activities, which are strongly related, happen periodically and must be performed by flexible systems from the standpoint of locomotion and data collection. Shafiekhani et al. (2017) [2], for example, proposes the Vinobot, a robot capable of collecting individualized plant data through a rod with temperature, humidity, and luminosity sensors and a robotic arm with an RGB-D camera for capturing depth images. The rod and the robotic arm allow data collection at different times in the plant's life in a flexible way, adapting to the size of the plants. The works by Mueller-Sim et al. (2017) [21] and Bao et al. (2019) [22] also present robots with similar configurations, but with distinct sensor and flexibility for data collection. After collecting data in the field, in these works, post-processing takes place to organize and make the data available. In the case of systems that collect depth images, point clouds or 3D models can be used for structural analysis processes. The analysis of works and techniques on 3D reconstruction of plants, therefore, are important to print to the models generated the highest possible accuracy, ensuring reliability to the system.

#### 2.2. 3D Plants Reconstruction

With the development and cost reduction of 3D sensors, 3D reconstruction research has come under the attention of researchers. Engineering [34–36], architecture [37,38] and medicine [39] are some areas benefited by this technology. The basic process of 3D reconstruction can be described in the following steps: data acquisition; point cloud preprocessing; point cloud fusion, when sequential depth images or images from different

sources are used; object model generation or update; and surface reconstruction [40]. Three-dimensional reconstruction can be classified in two ways: rigid and non-rigid 3D reconstruction. Both are optimization problems, where one seeks to minimize the error during the merging of the point clouds. However, in rigid 3D reconstruction, the objects in the scene are static, while in non-rigid 3D reconstruction, these objects are dynamic and have some level of movement [40–42]. The Iterative Closest Point (ICP) algorithm, proposed by Besl and McKay (1992) [41], is already well established for rigid 3D reconstructions, and several other works have augmented and improved it [43–45]. As for non-rigid 3D reconstruction, there is still no single strongly consolidated technique, but some works [46–52], such as DynamicFusion [46] and Fusion4D [48], have proposed techniques for model deformation that are shown to be very efficient.

The 3D reconstruction systems for agriculture, especially those aimed at creating 3D models of plants, can be classified into those for indoor environments (controlled) and those for open environments, using images from the field. Naturally, indoor systems have higher accuracy and quality compared to outdoor systems, since the images can be collected in a standardized way and without the influence of external elements such as rain and wind. In these systems, the processing flow of the depth images follows the traditional and some techniques seem to be trending. To perform point cloud fusion, when the case, ICP [53] and Coherent Point Drift (CPD) [7] techniques were used; for the reconstruction of the model surface, triangulated mesh generation techniques [3,6,7,54] and Non-uniform Rational B-Spline (NURBS) representation were mostly used [53,55]. Open environment oriented systems, on the other hand, are recent and are in early stages [2,22,56]. They basically use single depth images to reconstruct the 3D models of the plants, only treating the point clouds, and aim to generate accurate models for taking structural measurements. However, these systems share the double difficulty of proposing a data collection system and a data processing system for future use. Both indoor and outdoor systems share two characteristics: they aim to provide inputs for phenotyping processes; and they perform rigid 3D reconstruction of plants. From the analysis of these works, it was identified the absence of non-rigid 3D reconstruction in agricultural systems, an important process to allow natural characteristics, such as wind, to be considered in the 3D reconstruction; this type of reconstruction is explored in this work.

## 3. Proposed System: Materials and Methods

Considering the objective of this work: to use images and field data to perform a non-rigid 3D reconstruction of the plants and provide inputs for the phenotyping process, the proposed system was structured to collect the data and then process them. Therefore, its development was divided into two modules, specifically designed for these activities. The terrestrial acquisition module focuses mainly on the hardware components of the system, such as the robot, the sensors, and the RGB-D camera. The processing module is composed of the software responsible for generating the 3D model of the plant and merging this model with the sensor data. The integration between these modules happens through bag files [57]. The terrestrial acquisition module stores the data collected by the sensors in a bag file and the RGB-D videos captured in another bag file; the processing module, in turn, reads these files and performs the necessary processing in order to generate the 3D model of the plant and represent the sensor data in this model through its colorization. Next, the components of each of the modules, the development architectures, and the details of the techniques involved in the depth images processing will be presented.

#### 3.1. Robotic System for Terrestrial Data Acquisition

Integrating and managing various components of a robotic system is a complex task. The Robot Operating System (ROS) offers an architecture and features that facilitate the integration and control of the components of a robot. Thus, the terrestrial acquisition module was developed having as base architecture the ROS architecture [57,58]. From

this definition, the module's equipment and the nodes that compose the system could be defined and developed.

The main component of the module is the Jackal robot (unmanned ground vehicle), developed by Clearpath Robotics [59]. This robot has actuators for navigation, an on-board computer, GPS, LiDAR, Inertial Measurement Unit (IMU), and a 32-bit microcontroller for motor control, all integrated into a ROS. The ROS master runs on the robot's on-board computer, which has a collection of native nodes and topics to read data and control its equipment. Through the ROS architecture, it is possible to integrate custom equipment into the robot; either by creating external nodes and establishing communication with the native nodes or by connecting equipment to the on-board computer's Universal Serial Bus (USB) ports and creating internal nodes to control this equipment.

Carefully analyzing the problem of monitoring a corn plantation, some variables that are important to follow its growth were defined; as the size of the plant varies over time, these variables should be followed at different heights. Thus, a 1.5 m aluminum rod was installed in the robot, with supports for installing the sensors. The supports can be positioned at different heights, allowing more flexibility for data collection according to the life moment of the plantation. The variables chosen to monitor the plants were: temperature, humidity, and luminosity. Through these variables, it is possible to evaluate the environmental conditions in which the plantation is inserted, allowing the phenotyping process to define actions to solve problems in plant growth. Besides the sensor supports, the rod also allows the installation of an RGB-D camera, at different heights. Through the depth images, it is possible to reconstruct 3D models of the plants. Figure 2 shows the robot with the rod, the RGB-D camera installed and the sensor support in detail.



**Figure 2.** Proposed terrestrial acquisition module. (**a**) Overview; (**b**) RGB-D camera; and (**c**) sensor support.

The choice of sensors and RGB-D camera took into account the accuracy of data collection, the equipment size and the price. The RGB-D camera chosen was a stereo depth camera, the Intel RealSense D435i [60]; this camera has sensors capable of capturing depth images in low light environments, being ideal for captures in closed plantation [60,61]. The camera was connected directly to the robot's on-board computer via USB. For the temperature and humidity acquisition, the HDC1080 [62] sensor was used; this sensor has a resolution of 14 bits, providing accuracy of  $\pm 2\%$  for relative humidity and  $\pm 0.2$  °C for

temperature and has measurement time of approximately 6.50 ms for relative humidity and 6.35 ms for temperature. For luminosity acquisition, the BH1750FVI [63] sensor was used, with 16 bit resolution, maximum precision of  $\pm 20\%$  (lux) and measurement time of approximately 120 ms. The data collected by the sensors are read, using the Inter-Integrated Circuit (I<sup>2</sup>C) communication protocol, and processed by an Arduino MEGA 2560 [64]; and, then, transmitted to the requesting nodes via ROS service. Data can be collected by synchronizing the sensors measurement times with the desired video time, allowing numerous collection configurations. The proposed system was set up to collect two seconds of video and only one measurement from each sensor for each plant. Three sets of sensors were installed on the rod and all cabling for sensor data transmission was mounted with Unshielded Twisted Pair (UTP) cables. The Arduino is powered by and communicates serially (*rosserial*) with the robot's on-board computer also via the USB port. The collected data is stored in bag files that are read by the processing module to generate the 3D model of the plant and merge this model with the sensors data.

#### 3.2. Non-Rigid 3D Plant Reconstruction System

The terrestrial acquisition module, described previously, generates data that serves as input to the processing module. The goal of this processing is to perform the 3D reconstruction of the object present in the depth video, by means of techniques capable of processing non-rigid elements in the scene. The processing flow resembles that presented by Newcombe, Fox, and Seitz (2015) [46] and others that follow the same processing line. The system was developed in C++ with libraries for 3D object processing, among them: Intel RealSense SDK 2.0 [65], Point Cloud Library (PCL) [66], Computational Geometry Algorithms Library (CGAL) [67] and cpu\_tsdf Library [68].

The first operation of the system is to read the bag files generated by the terrestrial acquisition module. From these files, the frames of the depth videos and the data recorded by the sensors are extracted. Then, from those depth images, the input point cloud is generated. This procedure involves the removal of the background, using a novel technique based on the logarithmic function; standardization of the point cloud, using the octree data structure; removal of outliers using the average distance between points; and smoothing the points position using the Bilateral Smoothing technique. With the point cloud presenting as few problems as possible the input surface is reconstructed using the Advancing Front technique, based on the Delaunay triangulation. The first processed frame generates the first model, which is deformed at each iteration with a new input frame. For each iteration, the model surface is deformed to align with the input surface, also accumulating data. This operation involves describing the objects keypoints with the Signature of Histograms of Orientations (SHOT) technique; matching the keypoints (kd-tree search); deforming the model, using the Smoothed Rotation Enhanced As-Rigid-As Possible (SR-ARAP) technique; merging the surfaces, by merging the volumes described by the Truncated Signed Distance Function (TSDF); and further processing the model, removing new outliers, smoothing the points, and reconstructing the surface isotropically. After processing all the frames (key frames), the system merges this model with the data collected by the sensors through colorization. The colorized 3D model is then exported as a Polygon File Format (PLY) file [69], which can be effectively used as input for the phenotyping process. Figure 3 shows the flow described in graphical form. Further details of the operations mentioned will be presented next.



Figure 3. Proposed system processing flow.

#### 3.2.1. Background Removal

The point clouds generated from the depth frames carry a very large volume of data; for the standard resolution of the camera used ( $640 \times 480$  pixels) the generated point cloud has 307,200 points that mostly represent the scene background. The objective of removing these points is to reduce the volume of data processed by the other operations of the system and improve the model accuracy.

The images collected in the field have characteristics that make it difficult to segment, and therefore to remove the background, using only RGB (Red, Green and Blue) images; the main ones are the similarity of colors and shapes of the plants, even though they are at different distances. For this reason, only the collected depth data were used. To perform the background removal operation, a new technique was proposed, based on the logarithmic function, which proved to be more efficient than the linear background removal. This new technique uses a percentage cutoff factor to define the maximum depth value that will be kept in the point cloud. However, this value is found by relating this percentage and the minimum  $d_{min}$  and maximum  $d_{max}$  depth distances captured by the camera to the behavior of the logarithmic function.

The logarithmic function presents a behavior that can be efficiently used in positioning problems. Given the logarithmic function  $f(x) = \log_b x$  and considering f(x) represented in the ordinates axis and x represented in the abscissa axis; if a comparison with the linear function is done, it can be noticed that the logarithmic function curve presents a smooth transition from its values in the abscissa axis to a large portion of the ordinates axis values. This behavior changes, at a certain point of the ordinate values, becoming more aggressive in the abscissa values. This relationship between ordinates and abscissa varies according to the base b of the function. For the logarithmic function curve behavior to be used in background removal, it is necessary to relate the desired cutoff factor and depth distance references to the properties of this function. Two properties of the logarithmic function

are important to establish this relationship: the first is that  $\log_b 1 = 0$ ; the second is that  $\log_b b = 1$ . These properties can be used to convert percentages [0, 1] to values ranging from [1, b], since  $[\log_b 1, \log_b b]$  corresponds to the values [0, 1]. Thus, a background cutoff factor  $f_{bg}$  can be found by the equation  $f_{bg} = \log_b x_b$ , where  $x_b$  represents the cutoff distance in the logarithmic domain, found by linear conversion between [1, b] and  $[d_{min}, d_{max}]$ . Therefore, to find a depth value given a cutoff factor, it is sufficient to perform the reverse path, i.e., the cutoff factor  $f_{bg}$  must be converted by means of the exponential function to the interval [1, b] and linearly relate it to the interval  $[d_{min}, d_{max}]$ . For this case, the equation  $x_b = b^{f_{bg}}$  is sufficient to convert the cutoff factor  $f_{bg}$  to the interval [1, b]; and the linear conversion between the intervals [1, b] and  $[d_{min}, d_{max}]$  can be done with the equation

$$\frac{b-x_b}{b-1} = \frac{d_{max} - d_{bg}}{d_{max} - d_{min}} \,.$$
 (1)

Developing this equation and considering  $x_b = b^{f_{bg}}$ , the absolute depth distance value  $d_{bg}$ , which corresponds to the point cloud background, is found with the equation

$$d_{bg} = -\left(\frac{b - b^{f_{bg}}}{b - 1}\right) \cdot \Delta d + d_{max} ; \qquad (2)$$

where  $\Delta d = d_{max} - d_{min}$ . Any depth value greater than  $d_{bg}$  is removed from the input point cloud.

In practice, the proposed technique is a pass-through filter; however, the value that defines what is background or not is found using the logarithmic function. By the characteristic of this function, objects close to the camera can be prioritized, since it is possible to control the curve of the function by the logarithmic base. Furthermore, the proposed technique is less sensitive to the variation of minimum and maximum camera depths than the linear form, and remains effective for many scenarios of this variation. To aid the understanding of the proposed technique the Figures 4 and 5 illustrate the theoretical removal of the background and a practical example of this removal. In Figure 4, it is possible to observe the smoother variation of the proposed technique; the linear removal using a cutoff factor of 50%, always cuts the background in the middle between the minimum and maximum distances captured by the camera, whereas the proposed technique varies closer to the minimum distance. This is the desired behavior, since it was considered that the main object of the scene is closest to the camera. In Figure 5, it is possible to see the variation of the background removal using different logarithmic bases. After the initial adjustment of the cutoff factor, the proposed technique is able to remove the background dynamically and efficiently.



**Figure 4.** Examples of background removal for two logarithmic bases (10 and 100) of the proposed technique at different distances, compared to the linear form (lin). (a) Considering a cutoff factor of 0.4 and (b) considering a cutoff factor of 0.5.



**Figure 5.** Example of background removal in a point cloud considering a cutoff factor of 0.2. (**a**) Input point cloud ( $d_{min} = 0.29$  m and  $d_{max} = 22.9$  m); (**b**) linear cutting (at 4.81 m); (**c**) cutting by the proposed technique with base 10 (at 1.76 m); (**d**) cutting by the proposed technique with base 100 (at 0.63 m).

## 3.2.2. Point Cloud Standardization

The point cloud standardization process was also implemented to enable the control of the cloud data volume. This standardization allows adjusting the cloud resolution in a structured way, allowing to modify its accuracy according to the desired object in the scene. For the database studied, objects with large proportions can be described with a lower resolution than objects with smaller proportions without loss in model quality; i.e., at the beginning of the plant's life cycle, the point clouds should present a higher resolution for model generation, whereas for adult plants this resolution can be lower.

To perform this operation, the octree data structure was used to adaptively group sets of cloud points into voxels, according to a predefined resolution. For each of the voxels created, a single point describes the group [66]. This data consolidation can happen in four different ways: centroid of the points belonging to the voxel; centroid of the voxel position; first point inserted into the voxel; and last point inserted into the voxel. The first two ways are well known, but the last two are new and can be used to prioritize points in a cloud, in operations involving data from different sources, such as multiple cameras. In the tests, the proposed system used the centroid of the points belonging to the voxel to describe it. Figure 6 presents two examples of standardization, with 3 mm and 5 mm of resolution.



**Figure 6.** Point cloud standardization. (**a**) Input point cloud; (**b**) point cloud with 3 mm resolution and (**c**) point cloud with 5 mm resolution.

#### 3.2.3. Outliers Removal

Removing the background and standardizing the point cloud are operations that allow the control of data volume to process. However, either due to noise or the presence of unwanted elements in the scene, such as small pieces of leaves from other plants, scattered points may be present in the depth data. These points are undesirable, since their presence can compromise the accuracy of the model surface reconstruction and negatively influence the fusion operation of the input and model surfaces. For this reason, the outlier removal operation was implemented in the system.

This operation aims to remove points that have a distance among their neighbors outside the cloud pattern. To find these points, the average distance between points in the cloud must be calculated. Given a point cloud with M points and setting N neighbors, the average distance among points in the cloud  $d_m$  can be calculated [70] by the equation

$$d_m = \frac{1}{M} \sum_{i=1}^{M} \left( \frac{1}{N} \sum_{j=1}^{N} \sqrt{\left(x_i - x_j\right)^2 + \left(y_i - y_j\right)^2 + \left(z_i - z_j\right)^2} \right).$$
(3)

Then, for each point in the cloud, the distance between this point and its neighbors  $d_{local}$  is calculated and compared to the average distance  $d_m$  of the points in the cloud times a factor  $f_{out}$  (parameter), for example, twice the average distance ( $f_{out} = 2$ ). If  $d_{local} > d_m \cdot f_{out}$  the analyzed point is considered an outlier and removed from the point cloud. Figure 7 presets an example of outliers removal for  $f_{out} = 2$ , where circles indicate regions of outliers.



**Figure 7.** Outliers removal. (a) Input point cloud, the red circles indicate regions of outliers and (b) point cloud after the outliers removal operation ( $f_{out} = 2$ ).

#### 3.2.4. Point Cloud Smoothing

Similarly to the outlier removal operation, the smoothing of the cloud points position was implemented to improve the surface reconstruction quality. When collecting depth data, the camera may add small ripples to the description of originally flat regions, a factor caused by inaccuracies in the depth data capture. This feature contributes to a decrease in the surface reconstruction quality and description of keypoints in the surface fusion operation. The smoothing of the cloud points corrects these imperfections, providing greater accuracy in subsequent operations and in the generated model. To perform this operation the Bilateral Smoothing technique was used, composed of two parts: smoothing normals; and points repositioning based on the adjusted normals.

The Bilateral Smoothing technique needs the normal vectors to the implicit surface of the cloud points and its neighbors. These vectors are fitted, minimizing the distance to their neighbors, to describe smooth surfaces. The points are then positioned to fit the normals, so that they also form smooth surfaces. The estimation of the normal vector for each point is performed using the Principal Component Analysis (PCA) technique, which uses the eigenvalues and eigenvectors of the point covariance matrix. The smallest eigenvectors correspond to the best normal vectors with respect to the plane defined by the neighboring points, and the vector of this group with the smallest eigenvalue has the smallest direction variation and is perpendicular to the directions of greatest variation, corresponding to the normal vector to the surface [66].

The normals are iteratively adjusted, taking as a parameter the difference in distance from neighboring normals. The goal is to minimize these distances so that the discontinuities remain in separate groups, making continuous surfaces uniform and keeping the edges of the object [71]. After updating the normals, the cloud points are repositioned using the Locally Optimal Projector (LOP) technique proposed by Lipman et al. (2007) [72] and modified by Huang et al. (2013) [71] for the context of the Bilateral Smoothing technique. LOP redistributes a set of points in a way that adheres to the implicit shape described by those points. The smoothing operation can be done iteratively, allowing the control of the smoothing level. Figure 8 presents three examples of point cloud smoothing with different iteration numbers.



**Figure 8.** Point cloud smoothing. The orange point cloud indicates the smoothed point cloud, with (a) one iteration; (b) two iterations and (c) three iterations.

### 3.2.5. Advancing Front Surface Reconstruction

There are two forms of surface description of the objects processed by the proposed system: triangulated mesh (explicit representation), used to generate the deformation graph and the model; and TSDF volume (implicit representation), used to perform the fusion between the input surface and the model. The technique used to reconstruct the triangulated mesh was the Advancing Front [73,74]. This technique is based on the Delaunay triangulation, but has higher performance in terms of accuracy and quality, since the best triangles are formed sequentially. Another advantage of this technique is the ability to fill holes without having to perform this operation as preprocessing.

In the Advancing Front algorithm the initial triangle for mesh processing is defined by the size of the radius of the Delaunay triangles, defined by the circle passing over the three vertices of the triangle; the triangle with the smallest radius serves as the starting point of the algorithm. Then, the algorithm creates and maintains a list of candidate triangles to integrate the reconstructed surface. The list update happens every time a new triangle is incorporated into the partial surface, since new edges are made available for the reconstruction to move forward. For this list formation, there are four possibilities for a triangle to be a candidate to compose the surface: by extension; hole filling; ear filling; and glueing [73,74].

From the list of candidates, the triangle that has the highest plausibility degree is incorporated into the partial surface. The algorithm performs the update and definition of new triangles until all edges are verified. If a discontinuity occurs, the system performs the radius verification of the Delaunay triangles again, starting from the triangle with the smallest radius [73,74]. Figure 9 presents an example of surface reconstruction using the Advancing Front technique.



**Figure 9.** Advancing Front surface reconstruction. (**a**) Input point cloud and (**b**) surface reconstructed with the Advancing Front technique.

## 3.2.6. Keypoints Definition, Description and Correspondence

The model deformation operation, for the model and input surfaces fusion, needs the new position data for the vertices considered key, or keypoints. For this, implementations were necessary to define these vertices, describe them and find the pairs of similar vertices between the surfaces. This whole process, involving the keypoints, happens by means of a deformation graph. The generation of this graph happens by refining the object's surface, defining a new triangulated structure of lower resolution, but which faithfully describes the original surface. In the proposed system, the Incremental Triangle-based Isotropic Remeshing Algorithm [75] was implemented to generate the deformation graph. This algorithm performs simple operations incrementally on the surface edges and vertices; based on a predefined edge size, splitting, collapsing, inversion, as well as Laplacian smoothing (tangential projection) operations are performed. The operations are performed to describe a graph with triangles close to equilateral triangles; the Laplacian smoothing operation, in turn, seeks to adjust the triangles so the surface of the deformation graph is as close as possible to the original surface.

All vertices of the deformation graph mesh are considered keypoints. This process is performed for the input and model surfaces; and a mapping process is performed to maintain a relationship between the keypoints of the deformation graphs and the vertices of the source surface. With the keypoints defined the system is able to process the description and correspondence between these vertices.

In order for the model deformation operation to take place properly, the system must define the target position of the keypoints of the model deformation graph as best as possible. This definition is possible by matching the features of the model's keypoints and the input keypoints. Features such as position, topology, and neighborhood can be used to quantitatively describe a vertex of a triangulated mesh. Subsequently, these features can be compared, establishing a relationship of closeness and correspondence. By finding these correspondences, the target position of the model's deformation graph keypoints can be defined as the positions of their pairs in the input surface deformation graph. The description of the keypoints was implemented using the Signature of Histograms of Orientations (SHOT) technique [76].

Traditional 3D descriptors can be divided into two groups: descriptors by signature and descriptors by histograms. The first ones describe vertices by analyzing their neighborhood (support), defining an invariant reference frame, and encoding geometric measures of this support. The second ones describe the support by accumulating local geometric and topological measures, such as triangles areas and vertex counts in histograms. The SHOT descriptor brings these two forms of description together to create a stronger descriptor. With an isotropic spherical grid as the support structure, first-order entity histograms, such as the normals of the vertices in the support, and geometric information, such as the position of vertices, are extracted [76]. This support is sectioned along the radial, azimuth, and elevation axes; for each spatial division, a local histogram is computed. The construction of each local histogram takes into account the number of vertices and the value of the function that relates the angle between the normal of each vertex of the division and the normal of the keypoint.

To find the correspondence between the vertices of the deformation graphs, the search was implemented using the *k*-dimensional tree data structure (kd-tree). This structure is a specialization of the binary tree and allows for efficient search among nearest neighbors [66]. The *k* dimension used in the search is the SHOT descriptor of each vertex. The search process implemented in the proposed system was two-way search, improving the matching accuracy. In this process, first the vertices of the model deformation graph are compared with the vertices of the input deformation graph, then the input vertices are compared with those of the model; the matching pairs of vertices present in the two searches are defined as matching pairs. Figure 10 presents two examples of regions where keypoints are related, considering a sequence of depth images; the black lines indicate the match between vertices.



**Figure 10.** Keypoints correspondence. (a) Plant's stem region, comparison of deformation graphs between frames 30 (green) and 40 (red) and (b) plant's leaf region, comparison of deformation graphs between frames 40 (green) and 50 (red).

#### 3.2.7. Model Deformation

The keypoints definition, description, and matching between deformation graphs allow the matching points to form pairs of source and target positions for the model deformation process. Since the surface model is the one to be deformed, a proximity mapping was implemented between the vertices of this surface and the vertices of its deformation graph. Thus, the keypoints of the model surface are given target positions, making the model suitable to be deformed.

The technique implemented to perform the model deformation was the Smoothed Rotation Enhanced As-Rigid-As Possible (SR-ARAP) technique [77], a specialization of the As-Rigid-As Possible (ARAP) technique [78]. The deformation algorithm is similar to the ARAP deformation; however, a vertex rotation evaluation component is added to the energy function, providing greater transition smoothness in regions with steeper deformation. This technique is based on the Laplacian warping technique and uses a weighting scheme between mesh edges to drive the movement of vertices that are not keypoints.

In the deformation process, the surface keypoints are considered control vertices, since their movement influences the positioning of the other vertices, which compose the deformation region of interest. This movement propagation is accomplished by means of weights for the edges, which define what will be the movement portion of the vertex in relation to the moved control vertex. For the proposed system, the cotangent weighting scheme was implemented, which defines the weight by the ratio between the angles opposite to a given edge. Given the angles  $\theta_A$  and  $\theta_B$  opposite to an edge  $e_i$ , the weight of that edge  $w_{e_i}$  is given by Equation [79]

$$w_{e_i} = \cot(\theta_A) + \cot(\theta_B) . \tag{4}$$

Minimizing an energy function E(S') leads to an optimal deformation; which converges, with small error, the S surface to a deformed S' surface, considering the deformation constraints. The SR-ARAP technique introduces a bend element to the energy function; the addition of this term allows a control vertex and its neighborhood to obtain an optimal rotation in the deformation process. Thus, given a surface S with n vertices  $\mathbf{v}_i i \in \{1...n\}$  and k control vertices, the energy function defining the SR-ARAP deformation is given by Equation [77,79]

$$E(\mathcal{S}') = \sum_{\mathbf{v}_i \in \mathcal{S}} \sum_{\mathbf{v}_j \in N(\mathbf{v}_i)} w_{ij} \left\| (\mathbf{v}'_i - \mathbf{v}'_j) - \mathbf{R}_i (\mathbf{v}_i - \mathbf{v}_j) \right\|^2 + \alpha A \left\| \mathbf{R}_i - \mathbf{R}_j \right\|_F^2;$$
(5)

 $\mathbf{R}_i$  is the optimal 3 × 3 rotation matrix from source vertex position to target position;  $w_{ii}$ represents the edge weight;  $N(\mathbf{v}_i)$  represents the set of vertices adjacent to  $\mathbf{v}_i$  in  $\mathcal{S}$ ;  $\alpha A$  is the scalar weighting, where A represents the surface area to make the energy invariant to global scaling; F indicates that the difference between the rotation matrices of the vertices and their neighbors is calculated by the Frobenius matrix norm; and  $N(\mathbf{v}_i)$  represents the new position of the vertices  $N(\mathbf{v}_i)$  after deformation. The minimization of this equation can be simplified to the minimization of the energy of each vertex  $\mathbf{v}_i$ , considering that there are two unknowns: the new positions of vertices that are not part of the set of control vertices; and the rotation matrices  $\mathbf{R}_i$ . This minimization can be performed in two steps (local/global method) and iteratively, to further improve the deformation result [77–79]. In the proposed system, two stopping parameters were set for the deformation algorithm; either a pre-defined maximum number of iterations or a minimum value of the energy function E(S'). The Figures 11 and 12 present examples of surface deformation in two different regions of the plant, considering a sequence of depth images; in these figures it is possible to visualize how the proposed system performs the non-rigid deformations to align the surfaces for subsequent fusion.



**Figure 11.** Model deformation in plant's stem region. (a) Difference of the plant's stem position between the model deformation graph (green) and the input surface deformation graph (red), during processing between frames 30 and 40; (b) deformation of the model surface and (c) comparison between deformed model surface (green) and input surface (red).



**Figure 12.** Model deformation in plant's leaf region. (a) Difference of the plant's leaf position between the model deformation graph (green) and the input surface deformation graph (red), during processing between frames 40 and 50; (b) deformation of the model surface and (c) comparison between deformed model surface (green) and input surface (red).

## 3.2.8. Input and Model Surfaces Fusion

The model deformation operation results in a surface aligned to the input surface of a given system iteration. However, these two surfaces have different areas; the model surface describes accumulated regions of the previously processed surfaces, while the input surface may contain new elements, arising from the object's movement. Therefore, in order for the model to describe these new regions it is necessary to update it, accumulating these new parts of the objects to the existing parts. Another important factor in this surface fusion is that this process is able to remove small gaps between them, arising from errors in the deformation process, consolidating the model into a surface free of duplicate regions.

In the proposed system, the fusion of the input and model surfaces happens by updating the Truncated Signed Distance Function (TSDF) [80] volumetric representation of the model with respect to the same input representation. This implicit form of surface

representation defines Signed Distance Function (SDF) values for the voxels of a uniform grid; where SDF values close to 0 represent the surface of the object. The SDF values can be truncated at  $\pm t$  (TSDF), removing values far from the object's surface and reducing the amount of data processed and memory usage. Through this representation, it is possible to perform volume fusion by weighted averaging of TSDF values [80,81].

For the surface fusion to be performed by means of TSDF volumes (implicit representation) it was necessary to implement conversion techniques between this form of representation and explicit representations (triangulated meshes). The conversion from explicit to implicit representation was performed by an adaptive grid using the octree data structure [68], where for each voxel, a TSDF value and a weight (uncertainty) were associated. To perform the conversion from the implicit representation to the explicit representation, the marching cubes technique was used [66,82].

The surfaces fusion always happens at the end of each input frame processing. Figure 13 presents an example of surfaces fusion; in this figure, it is possible to see the consolidation of the surfaces and the disappearance of small spaces between them. In an iterative way, this process allows the creation of a precise model with data accumulation over time. After processing the last frame, the model vertices are smoothed, new outliers are removed, and a new isometric surface is regenerated, thus standardizing the generated model.



**Figure 13.** Input and model surfaces fusion. (**a**) Overlap of the deformed model surface (green) and the input surface (red) and (**b**) new model after surfaces fusion.

### 3.2.9. Model and Sensors Data Fusion

After processing all the input frames and generating the plant model, the system performs the fusion of the data collected by the sensors with the model. This fusion happens through the colorization of the model regions, according to the sensor values of the heights where they were installed; in the case of the proposed system, three heights. This way of fusion allows a quick visualization of the environmental conditions in which the plant is inserted. The system implements the data fusion in a separate way; that is, *n* copies of the models are created, one for each variable collected. To perform the model colorization, the linear interpolation of the RGB color space channels was implemented, relating the sensor heights; the sensor values; and parametric values of limits, of the variables and their colors. It was considered the linear interpolation Equation [83]

$$c(x) = c_{min} + \frac{x - x_{min}}{x_{max} - x_{min}} \cdot (c_{max} - c_{min}) ;$$
 (6)

where c(x) represents the value of a given channel of the RGB color space, for a reference value x;  $c_{min}$  and  $c_{max}$ , the limit values of the color channel;  $x_{min}$  and  $x_{max}$ , the limit values of the reference value x.

Initially, the system defines the  $RGB_i$  values for each measurement of the  $s_i$  sensors, by means of Equation (6); taking into account the measurement and color limits for each variable (parameters). Thus, for each of the heights  $h_i$ , a  $RGB_i$  value is defined. The processing to define the color of each vertex on the surface model involves the values  $h_i$ ,  $RGB_i$  and the vertex position. The pairs ( $h_{i-1}$ ,  $h_i$ ) and ( $RGB_{i-1}$ ,  $RGB_i$ ) replace, in Equation (6), ( $x_{min}$ ,  $x_{max}$ ) and ( $c_{min}$ ,  $c_{max}$ ), respectively; and the reference value x is replaced by the y axis value of the vertex position. As a result, the model surface is gradient colored, with the sensor heights as stop points. To illustrate this data fusion, Figure 14 presents an example of model and temperature sensor data fusion. The next section presents some outputs of the proposed system as results, from model generation to sensor data fusion.



**Figure 14.** Model and temperature sensor data fusion. (**a**) Model and (**b**) colorized model with data from the temperature sensor.

### 4. Results

During the system development, several minor tests were conducted to verify the implemented functionalities and evaluate possible adjustments. Although these tests used images from plants belonging to the target database, new tests were conducted with the fully processing model for the 3D reconstruction implemented. From these tests, it was possible to collect data in the field and produce system outputs in a standardized way, which will be presented next.

First, the terrestrial acquisition module was configured and the camera was calibrated; then the camera and sensors were positioned on the rod. The camera was positioned at 1 m height from the ground without tilting, while the sensors were positioned at 0.35 m, 1.35 m and 1.70 m. After these procedures, the module was taken to the field, in a corn plantation, where the depth images and sensor data were collected. Thus, the depth camera was positioned about 40 cm from the plants and the sensors about 5 cm from the stems. For each plant, a two seconds video was recorded by the camera and one measurement from each sensor was collected. Data were collected both from inside and from outside of the plantation at two periods of the day; one collection sequence in the morning and one in the afternoon. In the morning the sky was completely cloudy, but in the afternoon the weather got better and the sky was clear with high incidence of sunlight on the plants. These weather conditions allowed to verify differences in data collection, especially in sensor values.

Before proceeding with data processing, a validation was performed on the collected depth data. Thus, it was possible to verify that the data used to generate the models were accurate and reliable, since these measurements can be used in phenotyping processes. Thus, static depth images were collected from 20 plants, so that the whole plant was

captured (camera at 3 m distance), then plant heights extracted from these images were compared with the heights measured manually. Figure 15 presents the scatter plot of the measurements. It can be seen that the scatter was low, resulting in a Root Mean Square Error (RMSE) of 0.022 m or, if normalized by range (NRMSE), 1.75% error. These performance metrics indicate that the collected depth images have good accuracy and can be used for model generation.



**Figure 15.** Scatter plot of plant height measurements; comparison between depth-based height and manual height measurements.

The images and data were then processed. The system was set up to process two seconds of video, equivalent to 180 frames; the models, however, were generated using the sequence of frames from frame 30 to frame 150, with a 10 frame interval, totaling 13 frames processed. This initial frame cutting was necessary due to the camera startup time, which causes image distortions; the final frame cutting was done to maintain the time symmetry of the data used. Figure 16 presents some examples of generated models. In this figure, it is possible to see, besides the final model, the optical image, the depth image and the main processing steps. It can also be seen that, by means of these models, it is possible to extract measurements such as leaf width and length, angle between leaves and stems, measurements of the corn cobs, etc. To show the ability of the system to reconstruct the model in a non-rigid way and accumulate data over time, Figure 17 presents the models overlap of the processed frames, as well as the initial model and the final generated model. In Figure 17a,b, it is possible to visualize the movement of the plant leaves, especially in the overlapping detail section; even with this movement, the system performs the necessary deformations in the model to generate a final model without the presence of multiple leaves or undue deformations. Figure 17c presents a sharp example of data accumulation over time; in the initial frame the partial model presents thin leaves and even the absence of leaves, such as the lower right leaf; after the merging sequence, the final model presents thicker leaves and the presence of the lower right leaf.

Figure 18, in turn, exemplifies the fusion of sensor data with the models. This figure shows four different cases in data collection, so it was possible to observe all possible variations of the collected data. The models (a) and (b) were generated with data collected at the same time of day, which presented cloudy skies; it is possible to observe that plant (a) presents lower temperature, higher humidity, and progressive illumination in comparison with plant (b). For being inside the row, the plant (a) is inserted in an environment that varies more; its base, which is in the middle of other plants, receives less illumination and heat, but retains more humidity, in relation to the top of the plant. This behavior is different from the plant outside the row, inserted in a more uniform environment. The plats (c) and (d) present the same relationship as the plants (a) and (b); however, the data collection took place at a time with greater incidence of sunlight, increasing the temperature and luminosity and reducing the humidity.



Figure 16. Examples of models generated by the proposed system.



**Figure 17.** Overlapping of the partial models along the processing for the generation of the plant model. (**a**–**c**) Examples of non-rigid deformation and data accumulation in the generated models.



**Figure 18.** Generated models and fusion with sensor data. (**a**) Plant inside the row with low light incidence (cloudy); (**b**) plant outside the row with low light incidence; (**c**) plant inside the row with high light incidence (sunny); and (**d**) plant outside the row with high light incidence.

#### 5. Discussion

The results presented in the previous section show that the system is able to accurately produce 3D models of non-rigid corn plants from images and field data. This feature is essential for the development of crop inspection techniques, especially when considering future applications of systems that reconstruct the crop as a whole and not just individual plants. In applications of this size, inherent characteristics of open environments must be considered; one of them is the constant movement of the crop, due to wind or rain. The models generated can be used as inputs for phenotyping processes or structural measurements, while the fusion of sensor data to the model allows a quick visualization of the data collected in the field and a quick evaluation of the environment in which the plant is inserted.

Despite the goal achieved and the contribution of this work in the area, some characteristics of the system can still be further explored and new techniques or improvements in existing techniques can be studied. Starting with the processing form; the proposed system was not developed for real-time applications, since its functionalities were developed to be executed in the Central Process Unit (CPU) using threads. This processing could be implemented with parallelism in the Graphics Processing Unit (GPU), significantly increasing the processing speed. The system was design for phenotyping purposes, which usually uses small plantation areas. With the system developed for GPU processing, a larger area could be analyzed. From the point of view of the implemented techniques, some of them could be re-evaluated for performance improvement, and others could be replaced by new techniques in future work: analyze other methods for outlier removal, trying to eliminate the step of calculating the distances between points; analyze other surface reconstruction methods, with the aim of increasing the accuracy of the model; improve the correspondence index between the deformation graphs vertices, since this correspondence directly influences the deformation quality of the model; study the possibility of using a logarithmic weight scheme between the edges of the model, to increase the performance of the deformation; add new terms to the energy function, such as a possible logarithmic term for positioning between vertices; perform tests throughout the plant's life cycle, not only in its adult phase; these are some examples of improvements for the proposed system.

Future works, using not only the concepts presented in this work, but consolidating other concepts from previous works, could be proposed for a broader exploration and inspection of the plantation: use of drones for data collection; use of mechanical arms for more dynamic collection of both data and images; fusion of aerial data (and images) and terrestrial data; new ways to merge data with 3D models, for example merging temperature and humidity data into the same color scheme; among others. The main motivation for the development of these future works is the importance that agricultural crops represent, from a food and goods production point of view. By improving the efficiency of inspection and monitoring of the plant's life cycle, production in the field should increase.

### 6. Conclusions

Considering the current and future needs of agriculture, this work presented a system for field data acquisition and, which we believe to be for the first time, a non-rigid 3D reconstruction system for plants, to provide inputs for phenotyping processes. Through the developed modules, the field data collection and the model generation provide to the producer more efficiency in the plantation inspection and some inputs for plant structural evaluation.

The terrestrial acquisition module developed was composed of a robot, to which sensors and a depth camera were attached; thus, it was possible to collect data in the field in a more efficient way when compared to manual collection. The processing module was developed by implementing efficient techniques, evaluated from previous works, as well as new ones, such as the background removal based on the logarithmic function. With high depth data collection accuracy (error less than 1.75%), it was possible to generate models from depth video and merge these models with sensor data. Future improvements were presented in order to provide the system with higher processing speed, higher accuracy in the models generation, and alternative ways of data fusion.

The present work and other related works certainly contribute to the introduction of new technologies in the field, providing greater production volume and cost reduction. Not only that, they also encourage innovative works that also look at the field, since the resources produced there are of utmost importance for food and to provide inputs for the processing industry.

Author Contributions: Conceptualization, G.S.S., L.A.S. and M.M.; methodology, G.S.S.; software, G.S.S.; validation, G.S.S., M.M.; formal analysis, G.S.S., L.A.S. and M.M.; investigation, G.S.S.; resources, G.S.S.; data curation, G.S.S.; writing—original draft preparation, G.S.S.; writing—review and editing, G.S.S., L.A.S. and M.M.; visualization, G.S.S.; supervision, L.A.S. and M.M.; project administration, L.A.S. and M.M.; funding acquisition, L.A.S. and M.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by Mackenzie Research Fund (MackPesquisa), grant number 191001.

**Acknowledgments:** This work was partially supported by the Mackenzie Research Fund (Mack-Pesquisa) at Mackenzie Presbyterian University. All tests were done in the farms of Christian Institute Technical High School, Castro, PR, Brazil.

**Conflicts of Interest:** The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

# Abbreviations

The following abbreviations are used in this manuscript:

3D	Three-dimensional
ARAP	As-Rigid-As-Possible
CGAL	Computational Geometry Algorithms Library
CPD	Coherent Point Drift
CPU	Central Process Unit
GPS	Global Positioning System
GPU	Graphics Processing Unit
ICP	Iterative Closest Point
I <sup>2</sup> C	Inter-Integrated Circuit
IMU	Inertial Measurement Unit
LiDAR	Light Detection Furthermore, Ranging
LOP	Locally Optimal Projector
NRMSE	Normalized Root Mean Square Error
NURBS	Non-uniform Rational B-Spline
PCA	Principal Component Analysis
PCL	Point Cloud Library
PLY	Polygon File Format
RGB	Red, Green and Blue
RGB-D	Red, Green, Blue and Depth
RMSE	Root Mean Square Error
ROS	Robot Operating System
SDF	Signed Distance Function
SHOT	Signature of Histograms of Orientations
SLAM	Simultaneous Localization and Mapping
SR-ARAP	Smoothed Rotation Enhanced As-Rigid-As Possible
TSDF	Truncated Signed Distance Function
USB	Universal Serial Bus
UTP	Unshielded Twisted Pair

### References

- FAOSTAT. Food and Agriculture Data from Food and Agriculture Organization of the United Nations. 2020 Available online: http://www.fao.org/faostat/en/#data (accessed on 27 January 2021).
- Shafiekhani, A.; Kadam, S.; Fritschi, F.; DeSouza, G. Vinobot and Vinoculer: Two Robotic Platforms for High-Throughput Field Phenotyping. Sensors 2017, 17, 214. [CrossRef]
- Kumar, P.; Connor, J.; Mikiavcic, S. High-throughput 3D reconstruction of plant shoots for phenotyping. In Proceedings of the 2014 13th International Conference on Control Automation Robotics & Vision (ICARCV), Marina Bay Sands, Singapore, 10–12 December 2014. [CrossRef]
- 4. Paulus, S.; Schumann, H.; Kuhlmann, H.; Léon, J. High-precision laser scanning system for capturing 3D plant architecture and analysing growth of cereal plants. *Biosyst. Eng.* **2014**, *121*, 1–11. [CrossRef]
- Schwambach, C.; Sobrinho, G.C. Biologia [Livro Eletrônico]; Coleção EJA: Cidadania Competente, v. 7; InterSaberes: Curitiba, Brazil, 2017; Chapter 7.
- Zhu, F.; Thapa, S.; Gao, T.; Ge, Y.; Walia, H.; Yu, H. 3D Reconstruction of Plant Leaves for High-Throughput Phenotyping. In Proceedings of the 2018 IEEE International Conference on Big Data (Big Data), Seattle, WA, USA, 10–13 December 2018. [CrossRef]
- Chaudhury, A.; Ward, C.; Talasaz, A.; Ivanov, A.G.; Brophy, M.; Grodzinski, B.; Huner, N.P.A.; Patel, R.V.; Barron, J.L. Machine Vision System for 3D Plant Phenotyping. *IEEE/ACM Trans. Comput. Biol. Bioinform.* 2019, 16, 2009–2022. [CrossRef]
- Potena, C.; Khanna, R.; Nieto, J.; Siegwart, R.; Nardi, D.; Pretto, A. AgriColMap: Aerial-Ground Collaborative 3D Mapping for Precision Farming. *IEEE Robot. Autom. Lett.* 2019, 4, 1085–1092. [CrossRef]
- 9. Weiss, U.; Biber, P. Plant detection and mapping for agricultural robots using a 3D LIDAR sensor. *Robot. Auton. Syst.* 2011, 59, 265–273. [CrossRef]
- 10. Mazoyer, M.; Roudart, L. *História das Agriculturas no Mundo: Do Neolítico à Crise Contemporânea*; UNESP-NEAD: São Paulo, Brazil, 2009; Chapters 1, 2, 9 and 10.
- 11. de Freitas Vian, C.E.; Júnior, A.M.A.; Baricelo, L.G.; da Silva, R.P. Origens, evolução e tendências da indústria de máquinas agrícolas. *Rev. Econ. Sociol. Rural.* **2013**, *51*, 719–744. [CrossRef]

- 12. Massruhá, S.M.F.S.; de Andrade Leite, M.A.; Junior, A.L.; Romani, L.A.S. *Tecnologias da Informação e Comunicação e suas Relações com a Agricultura*; Embrapa: Brasilia, Brazil, 2014; Chapter 1.
- 13. Duckett, T.; Pearson, S.; Blackmore, S.; Grieve, B.; Chen, W.H.; Cielniak, G.; Cleaversmith, J.; Dai, J.; Davis, S.; Fox, C.; et al. *Agricultural Robotics: The Future of Robotic Agriculture*; UK-RAS Network White Papers; UK Robotics and Autonomous Systems: London, UK, 2018.
- 14. CASE IH. Automation. 2020. Available online: https://www.caseih.com/northamerica/en-us/innovations/automation (accessed on 16 August 2020).
- 15. Agrointelli. Robotti. 2020. Available online: https://www.agrointelli.com/robotti (accessed on 16 August 2020).
- 16. Haibo, L.; Shuliang, D.; Zunmin, L.; Chuijie, Y. Study and Experiment on a Wheat Precision Seeding Robot. *J. Robot.* **2015**, 2015, 1–9. [CrossRef]
- 17. Srinivasan, N.; Prabhu, P.; Smruthi, S.S.; Sivaraman, N.V.; Gladwin, S.J.; Rajavel, R.; Natarajan, A.R. Design of an autonomous seed planting robot. In Proceedings of the 2016 IEEE Region 10 Humanitarian Technology Conference (R10-HTC), Agra, India, 21–23 December 2016. [CrossRef]
- Fendt. Fendt Puts the New Robot 'Xaver' to Use. 2017. Available online: https://www.fendt.com/int/fendt-xaver (accessed on 16 August 2020).
- 19. RISE ABOVE. Home Page. 2020. Available online: https://www.riseabove.com.au (accessed on 17 August 2020).
- 20. Ristorto, G.; Gallo, R.; Gasparetto, A.; Scalera, L.; Vidoni, R.; Mazzetto, F. A mobile laboratory for orchard health status monitoring in precision farming. *Chem. Eng. Trans.* 2017, *58*, 661–666. [CrossRef]
- Mueller-Sim, T.; Jenkins, M.; Abel, J.; Kantor, G. The Robotanist: A ground-based agricultural robot for high-throughput crop phenotyping. In Proceedings of the 2017 IEEE International Conference on Robotics and Automation (ICRA), Marina Bay Sands, Singapore, 29 May–3 June 2017. [CrossRef]
- Bao, Y.; Tang, L.; Srinivasan, S.; Schnable, P.S. Field-based architectural traits characterisation of maize plant using time-of-flight 3D imaging. *Biosyst. Eng.* 2019, 178, 86–101. [CrossRef]
- Chapman, S.; Merz, T.; Chan, A.; Jackway, P.; Hrabar, S.; Dreccer, M.; Holland, E.; Zheng, B.; Ling, T.; Jimenez-Berni, J. Pheno-Copter: A Low-Altitude, Autonomous Remote-Sensing Robotic Helicopter for High-Throughput Field-Based Phenotyping. *Agronomy* 2014, 4, 279–301. [CrossRef]
- Cantelli, L.; Bonaccorso, F.; Longo, D.; Melita, C.D.; Schillaci, G.; Muscato, G. A Small Versatile Electrical Robot for Autonomous Spraying in Agriculture. *AgriEngineering* 2019, 1, 391–402. [CrossRef]
- NAIO TECHNOLOGIES. DINO Autonomous Vegetable Weeding Robot—DINO. 2020. Available online: https://www.naio-tec hnologies.com/en/agricultural-equipment/large-scale-vegetable-weeding-robot (accessed on 17 August 2020).
- Ecorobotix. Technology for Environment. 2020. Available online: https://www.ecorobotix.com/en (accessed on 17 August 2020).
  DJI. A DJI Agriculture. 2020. Available online: https://ag.dji.com/pt-br (accessed on 20 August 2020).
- 28. BLUE RIVER TECHNOLOGY. Our See and Spray Machines. 2020. Available online: http://smartmachines.bluerivertechnology. com (accessed on 16 August 2020).
- 29. Silwal, A.; Davidson, J.R.; Karkee, M.; Mo, C.; Zhang, Q.; Lewis, K. Design, integration, and field evaluation of a robotic apple harvester. *J. Field Robot.* **2017**, *34*, 1140–1159. [CrossRef]
- 30. Agrobot. Agricultural Robots. 2020. Available online: https://www.agrobot.com (accessed on 17 August 2020).
- 31. Abundant Robotics. Home Page. 2020. Available online: https://www.abundantrobotics.com (accessed on 16 August 2020).
- 32. Energid. Citrus Harvesting. 2020. Available online: https://www.energid.com/industries/agricultural-robotics (accessed on 17 August 2020).
- 33. Oliveira, L.F.P.; Moreira, A.P.; Silva, M.F. Advances in Agriculture Robotics: A State-of-the-Art Review and Challenges Ahead. *Robotics* **2021**, *10*, 52. [CrossRef]
- 34. Sung, C.; Kim, P.Y. 3D terrain reconstruction of construction sites using a stereo camera. Autom. Constr. 2016, 64, 65–77. [CrossRef]
- 35. Wu, Q.; Yang, H.; Wei, M.; Remil, O.; Wang, B.; Wang, J. Automatic 3D reconstruction of electrical substation scene from LiDAR point cloud. *ISPRS J. Photogramm. Remote Sens.* **2018**, *143*, 57–71. [CrossRef]
- 36. Jing, N.B.; Ma, X.M.; Guo, W. 3D Reconstruction of Underground Tunnel Using Kinect Camera. In Proceedings of the 2018 International Symposium on Computer, Consumer and Control (IS3C), Taichung, Taiwan, 6–8 December 2018. [CrossRef]
- 37. Cheng, L.; Tong, L.; Chen, Y.; Zhang, W.; Shan, J.; Liu, Y.; Li, M. Integration of LiDAR data and optical multi-view images for 3D reconstruction of building roofs. *Opt. Lasers Eng.* **2013**, *51*, 493–502. [CrossRef]
- Hu, Y.; Chen, Y.; Wu, Z. Unmanned Aerial Vehicle and Ground Remote Sensing Applied in 3D Reconstruction of Historical Building Groups in Ancient Villages. In Proceedings of the 2018 Fifth International Workshop on Earth Observation and Remote Sensing Applications (EORSA), Xi'an, China, 18–20 June 2018. [CrossRef]
- Lacher, R.; Vasconcelos, F.; Williams, N.; Rindermann, G.; Hipwell, J.; Hawkes, D.; Stoyanov, D. Nonrigid reconstruction of 3D breast surfaces with a low-cost RGBD camera for surgical planning and aesthetic evaluation. *Med. Image Anal.* 2019, 53, 11–25. [CrossRef] [PubMed]
- 40. Zollhöfer, M.; Stotko, P.; Görlitz, A.; Theobalt, C.; Nießner, M.; Klein, R.; Kolb, A. State of the Art on 3D Reconstruction with RGB-D Cameras. *Comput. Graph. Forum* **2018**, *37*, 625–652. [CrossRef]
- 41. Besl, P.; McKay, N.D. A method for registration of 3-D shapes. IEEE Trans. Pattern Anal. Mach. Intell. 1992, 14, 239–256. [CrossRef]

- 42. Mora, H.; Mora-Pascual, J.M.; García-García, A.; Martínez-González, P. Computational Analysis of Distance Operators for the Iterative Closest Point Algorithm. *PLoS ONE* **2016**, *11*, e0164694. [CrossRef]
- 43. Henry, P.; Krainin, M.; Herbst, E.; Ren, X.; Fox, D. RGB-D mapping: Using Kinect-style depth cameras for dense 3D modeling of indoor environments. *Int. J. Robot. Res.* 2012, 31, 647–663. [CrossRef]
- 44. Huang, X.; Hu, M. 3D Reconstruction Based on Model Registration Using RANSAC-ICP Algorithm. In *Transactions on Edutainment XI*; Springer: Berlin/Heidelberg, Germany, 2015; Volume 8971, pp. 46–51. [CrossRef]
- 45. Newcombe, R.A.; Fitzgibbon, A.; Izadi, S.; Hilliges, O.; Molyneaux, D.; Kim, D.; Davison, A.J.; Kohi, P.; Shotton, J.; Hodges, S. KinectFusion: Real-time dense surface mapping and tracking. In Proceedings of the 2011 10th IEEE International Symposium on Mixed and Augmented Reality, Basel, Switzerland, 26–29 October 2011. [CrossRef]
- 46. Newcombe, R.A.; Fox, D.; Seitz, S.M. DynamicFusion: Reconstruction and tracking of non-rigid scenes in real-time. In Proceedings of the 2015 IEEE Conference on Computer Vision and Pattern Recognition (CVPR), Boston, MA, USA, 7–12 June 2015. [CrossRef]
- Innmann, M.; Zollhöfer, M.; Nießner, M.; Theobalt, C.; Stamminger, M. VolumeDeform: Real-Time Volumetric Non-rigid Reconstruction. In *Computer Vision – ECCV 2016*; Springer: Berlin/Heidelberg, Germany, 2016; Volume 9912, pp. 362–379. [CrossRef]
- 48. Dou, M.; Khamis, S.; Degtyarev, Y.; Davidson, P.; Fanello, S.R.; Kowdle, A.; Escolano, S.O.; Rhemann, C.; Kim, D.; Taylor, J.; et al. Fusion4D. *ACM Trans. Graph.* **2016**, *35*, 1–13. [CrossRef]
- 49. Dou, M.; Davidson, P.; Fanello, S.R.; Khamis, S.; Kowdle, A.; Rhemann, C.; Tankovich, V.; Izadi, S. Motion2fusion. *ACM Trans. Graph.* 2017, 36, 1–16. [CrossRef]
- Slavcheva, M.; Baust, M.; Cremers, D.; Ilic, S. KillingFusion: Non-rigid 3D Reconstruction without Correspondences. In Proceedings of the 2017 IEEE Conference on Computer Vision and Pattern Recognition (CVPR), Honolulu, HI, USA, 21–26 July 2017. [CrossRef]
- 51. Guo, K.; Xu, F.; Yu, T.; Liu, X.; Dai, Q.; Liu, Y. Real-Time Geometry, Albedo, and Motion Reconstruction Using a Single RGB-D Camera. *ACM Trans. Graph.* **2017**, *36*, 1–13. [CrossRef]
- Slavcheva, M.; Baust, M.; Ilic, S. SobolevFusion: 3D Reconstruction of Scenes Undergoing Free Non-rigid Motion. In Proceedings of the 2018 IEEE/CVF Conference on Computer Vision and Pattern Recognition, Salt Lake City, UT, USA, 18–23 June 2018. [CrossRef]
- 53. Chaivivatrakul, S.; Tang, L.; Dailey, M.N.; Nakarmi, A.D. Automatic morphological trait characterization for corn plants via 3D holographic reconstruction. *Comput. Electron. Agric.* **2014**, *109*, 109–123. [CrossRef]
- 54. Gibbs, J.A.; Pound, M.; French, A.; Wells, D.; Murchie, E.; Pridmore, T. Active Vision and Surface Reconstruction for 3D Plant Shoot Modelling. *IEEE/ACM Trans. Comput. Biol. Bioinform.* **2019**. [CrossRef]
- 55. Gelard, W.; Devy, M.; Herbulot, A.; Burger, P. Model-based Segmentation of 3D Point Clouds for Phenotyping Sunflower Plants. In Proceedings of the 12th International Joint Conference on Computer Vision, Imaging and Computer Graphics Theory and Applications, Porto, Portugal, 27 February–1 March 2017. [CrossRef]
- 56. Virlet, N.; Sabermanesh, K.; Sadeghi-Tehran, P.; Hawkesford, M.J. Field Scanalyzer: An automated robotic field phenotyping platform for detailed crop monitoring. *Funct. Plant Biol.* **2017**, *44*, 143. [CrossRef] [PubMed]
- 57. ROS. Documentation. 2020. Available online: http://wiki.ros.org (accessed on 19 September 2020).
- 58. Clearpath Robotics. Intro to ROS. 2015. Available online: http://www.clearpathrobotics.com/assets/guides/kinetic/ros/Intro% 20to%20the%20Robot%20Operating%20System.html (accessed on 20 September 2020).
- 59. Clearpath Robotics. Jackal Unmanned Ground Vehicle. 2020. Available online: https://clearpathrobotics.com/jackal-small-unm anned-ground-vehicle/ (accessed on 20 September 2020).
- 60. Intel Corporation. Intel<sup>®</sup> RealSense<sup>TM</sup> Product Family D400 Series; Intel Corporation: Santa Clara, CA, USA, 2020.
- 61. Intel Corporation. Intel RealSense Depth Camera D435i. 2020. Available online: https://www.intelrealsense.com/depth-camerad435i (accessed on 1 April 2020).
- 62. Texas Instruments. HDC1080 Low Power, High Accuracy Digital Humidity Sensor with Temperature Sensor; Texas Instruments Incorporated: Dallas, TX, USA, January 2016.
- 63. Rohm Semiconductor. *Digital 16bit Serial Output Type Ambient Light Sensor IC*; ROHM Semiconductor: Santa Clara, CA, USA, 2010.
- 64. Arduino. Arduino Mega 2560 Rev3. 2021. Available online: https://store.arduino.cc/usa/mega-2560-r3 (accessed on 30 March 2021).
- 65. Intel Corporation. Intel RealSense SDK 2.0. 2020. Available online: https://github.com/IntelRealSense/librealsense (accessed on 22 September 2020).
- 66. Rusu, R.B.; Cousins, S. 3D is here: Point Cloud Library (PCL). In Proceedings of the 2011 IEEE International Conference on Robotics and Automation (ICRA), Shanghai, China, 9–13 May 2011. [CrossRef]
- 67. The CGAL Project. *CGAL User and Reference Manual*. CGAL Editorial Board, 5.1 edition, 2020. Available online: https://doc.cgal.org/latest/Manual/how\_to\_cite\_cgal.html (accessed on 29 April 2021).
- 68. Miller, S. CPU TSDF. 2020. Available online: https://github.com/sdmiller/cpu\_tsdf (accessed on 22 September 2020).
- 69. Turk, G. The PLY Polygon File Format. 1994. Available online: https://web.archive.org/web/20161204152348/http://www.dcs. ed.ac.uk/teaching/cs4/www/graphics/Web/ply.html (accessed on 11 June 2021).

- Alliez, P.; Giraudot, S.; Jamin, C.; Lafarge, F.; Mérigot, Q.; Meyron, J.; Saboret, L.; Salman, N.; Wu, S.; Yildiran, N.F. Point Set Processing. In *CGAL User and Reference Manual*; CGAL 5.1; 2020. Available online: https://doc.cgal.org/latest/Point\_set\_processing\_3/index.html (accessed on 29 April 2021).
- Huang, H.; Wu, S.; Gong, M.; Cohen-Or, D.; Ascher, U.; Zhang, H.R. Edge-aware point set resampling. ACM Trans. Graph. 2013, 32, 1–12. [CrossRef]
- 72. Lipman, Y.; Cohen-Or, D.; Levin, D.; Tal-Ezer, H. Parameterization-free projection for geometry reconstruction. In ACM SIGGRAPH 2007 Papers on SIGGRAPH '07; ACM Press: San Diego, CA, USA, 2007. [CrossRef]
- 73. Cohen-Steiner, D.; Da, F. A greedy Delaunay-based surface reconstruction algorithm. Vis. Comput. 2004, 20, 4–16. [CrossRef]
- 74. Da, T.K.F.; Cohen-Steiner, D. Advancing Front Surface Reconstruction. In CGAL User and Reference Manual; CGAL 5.1; 2020. Available online: https://doc.cgal.org/latest/Advancing\_front\_surface\_reconstruction/group\_PkgAdvancingFrontSurface ReconstructionRef.html (accessed on 29 April 2021).
- 75. Botsch, M.; Kobbelt, L. A remeshing approach to multiresolution modeling. In Proceedings of the 2004 Eurographics/ACM SIGGRAPH Symposium on Geometry Processing—SGP '04, Nice, France, 8–10 July 2004. [CrossRef]
- Tombari, F.; Salti, S.; Stefano, L.D. Unique Signatures of Histograms for Local Surface Description. In *Computer Vision—ECCV* 2010; Springer: Berlin/Heidelberg, Germany, 2010; pp. 356–369. [CrossRef]
- 77. Levi, Z.; Gotsman, C. Smooth Rotation Enhanced As-Rigid-As-Possible Mesh Animation. *IEEE Trans. Vis. Comput. Graph.* **2015**, *21*, 264–277. [CrossRef]
- Sorkine, O.; Alexa, M. As-Rigid-As-Possible Surface Modeling. In Proceedings of the Fifth Eurographics Symposium on Geometry Processing (SGP '07), Barcelona, Spain, 4–6 July 2007; pp. 109–116. [CrossRef]
- Loriot, S.; Sorkine-Hornung, O.; Xu, Y.; Yaz, I.O. Triangulated Surface Mesh Deformation. In CGAL User and Reference Manual; CGAL 5.1; 2020. Available online: https://doc.cgal.org/latest/Surface\_mesh\_deformation/index.html (accessed on 29 April 2021).
- Curless, B.; Levoy, M. A volumetric method for building complex models from range images. In Proceedings of the 23rd Annual Conference on Computer Graphics and Interactive Techniques—SIGGRAPH '96, New Orleans, LA, USA, 4–9 August 1996. [CrossRef]
- 81. Werner, D.; Al-Hamadi, A.; Werner, P. Truncated Signed Distance Function: Experiments on Voxel Size. In *Lecture Notes in Computer Science*; Springer: Berlin/Heidelberg, Germany, 2014; Volume 8815, pp. 357–364. [CrossRef]
- Lorensen, W.E.; Cline, H.E. Marching cubes: A high resolution 3D surface construction algorithm. In Proceedings of the 14th Annual Conference on Computer Graphics and Interactive Techniques—SIGGRAPH '87, Anaheim, CA, USA, 27–31 July 1987, [CrossRef]
- 83. Gautschi, W. Approximation and Interpolation. In Numerical Analysis; Birkhäuser: Boston, MA, USA, 2011; pp. 55–158. [CrossRef]